## Check for updates

## **OPEN ACCESS**

EDITED BY Muthuraman Muthuraman, Johannes Gutenberg University Mainz, Germany

REVIEWED BY Nelleke van Wouwe, University of Louisville, United States Qichang Mei, Ningbo University, China

\*CORRESPONDENCE Joshua K. Wong Joshua.wong@neurology.ufl.edu

<sup>†</sup>These authors have contributed equally to this work and share first authorship

<sup>‡</sup>These authors share senior authorship

#### SPECIALTY SECTION

This article was submitted to Brain Imaging and Stimulation, a section of the journal Frontiers in Human Neuroscience

RECEIVED 19 July 2022 ACCEPTED 05 September 2022 PUBLISHED 29 September 2022

#### CITATION

King C, Parker TM, Roussos-Ross K, Ramirez-Zamora A, Smulian JC, Okun MS and Wong JK (2022) Safety of deep brain stimulation in pregnancy: A comprehensive review. *Front. Hum. Neurosci.* 16:997552. doi: 10.3389/fnhum.2022.997552

#### COPYRIGHT

© 2022 King, Parker, Roussos-Ross, Ramirez-Zamora, Smulian, Okun and Wong. This is an open-access article distributed under the terms of the **Creative Commons Attribution License** (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# Safety of deep brain stimulation in pregnancy: A comprehensive review

Caroline King<sup>1†</sup>, T. Maxwell Parker<sup>2†</sup>, Kay Roussos-Ross<sup>1,3</sup>, Adolfo Ramirez-Zamora<sup>2</sup>, John C. Smulian<sup>1,3‡</sup>, Michael S. Okun<sup>2‡</sup> and Joshua K. Wong<sup>2\*</sup>

<sup>1</sup>Department of Obstetrics & Gynecology, University of Florida College of Medicine, Gainesville, FL, United States, <sup>2</sup>Department of Neurology, Norman Fixel Institute for Neurological Diseases, University of Florida College of Medicine, Gainesville, FL, United States, <sup>3</sup>Center for Research in Perinatal Outcomes, University of Florida College of Medicine, Gainesville, FL, United States

**Introduction:** Deep brain stimulation (DBS) is increasingly used to treat the symptoms of various neurologic and psychiatric conditions. People can undergo the procedure during reproductive years but the safety of DBS in pregnancy remains relatively unknown given the paucity of published cases. We thus conducted a review of the literature to determine the state of current knowledge about DBS in pregnancy and to determine how eligibility criteria are approached in clinical trials with respect to pregnancy and the potential for pregnancy.

**Methods:** A literature review was conducted in EMBASE to identify articles involving DBS and pregnancy. Two reviewers independently analyzed the articles to confirm inclusion. Data extracted for analysis included conditions treated, complications at all stages of pregnancy, neonatal/pediatric outcomes, and DBS target. A second search was then conducted using www.clinicaltrials.gov. The same two reviewers then assessed whether each trial excluded pregnant individuals, lactating individuals, or persons of childbearing age planning to conceive. Also assessed was whether contraception had to be deemed adequate prior to enrollment.

**Results:** The literature search returned 681 articles. Following independent analysis and agreement of two reviewers, 8 pregnancy related DBS articles were included for analysis. These articles described 27 subjects, 29 pregnancies (2 with subsequent pregnancies), and 31 infants (2 twin pregnancies). There was 1 preterm birth at 35 weeks, and 3 patients who experienced discomfort from the DBS battery (i.e., impulse generator) placement site. All 27 patients had a DBS device implanted before they became pregnant, which remained in use throughout their pregnancy. There was exclusion of pregnant individuals from 68% of 135 interventional trials involving DBS. Approximately 44% of these trials excluded persons of childbearing age not on "adequate contraception" or wishing to conceive in the coming years. Finally, 22% excluded breastfeeding persons.

**Conclusion:** The data from 29 pregnancies receiving DBS treatment during pregnancy was not associated with unexpected pregnancy or post-partum

complication patterns. Many clinical trials have excluded pregnant individuals. Documentation of outcomes in larger numbers of pregnancies will help clarify the safety profile and will help guide study designs that will safely include pregnant patients.

KEYWORDS

pregnancy, DBS, clinical trials, safety, ethics, deep brain stimulation, neuromodulation

# Introduction

Deep brain stimulation (DBS) has been applied selectively for treatment of Parkinson's disease (PD), essential tremor (ET), dystonia, and many other neuropsychiatric disorders and symptoms (Denison and Morrell, 2022). The steadily improving safety profile has led to expansion into younger and healthier populations during reproductive ages, which raises important issues regarding DBS and pregnancy. Individuals with DBS may be interested in becoming pregnant or may unknowingly discover that they are pregnant after a DBS device has been implanted. Others may also be interested in the safety profile during lactation and whether or not they should consider enrollment in a clinical trial if pregnant or considering a future pregnancy.

There has been an increase in clinical trials utilizing DBS across 28 conditions, many of which are in the psychiatric or cognitive disease domains (Harmsen et al., 2020). Many focus on younger patients and thus pregnancy-related issues become more pertinent (Schrag and Schott, 2006; Louis and Dogu, 2008). With mean maternal age at first birth in the United States at 27 years, the use of DBS in pregnancy becomes an increasingly important topic (CDC, 2022). Additionally, many of the expanding indications for neuromodulation, such as major depressive disorder (MDD) and epilepsy, affect pregnant individuals. Each year, up to 20% of pregnant individuals suffer from a depressive disorder and over 1.1 million women of childbearing age have epilepsy (Sazgar, 2019; Van Niel and Payne, 2020). Uncontrolled and poorly controlled MDD and epilepsy in pregnancy pose significant threats to both mother and fetus, ranging from neurodevelopmental derangements to fetal hypoxia and growth restriction and even death (Battino and Tomson, 2007; Chan et al., 2014). In refractory cases of epilepsy (affecting 40% of all persons with the disease), seizures can increase in frequency during pregnancy (Kobau et al., 2008; Vitturi et al., 2019). Many first-line treatment modalities for these disorders, such as anti-epileptic pharmacotherapeutics, can cross the placenta and have known teratogenic effects on fetuses or neurodevelopmental delay during childhood, limiting the availability of safe and effective treatments in pregnancy (Artama et al., 2005). In these instances, the patient may benefit from alternative non-pharmacologic therapeutic approaches such as DBS.

It is thus an imperative to assess and analyze the data available surrounding DBS and the pregnant population with the most up to date evidence. Doing so will shed light on crucial areas of promise and progress, as well as better characterize considerations for each neurologist and obstetrician's own practice. In this comprehensive review we will address pregnancy-related DBS concerns including clinical trial enrollment using all available studies in the published literature. Our primary objective was to describe clinical outcomes of pregnancies in which DBS was used. Our secondary objective was to assess how clinical trials approached participation eligibility during pregnancy, lactation, and the reproductive years.

## Methods

A comprehensive review of the literature for DBS in the childbearing and pregnant populations was conducted from March 2022 to June 2022 querying EMBASE for all cases reported to date. Our search criteria is available in Supplementary Data Sheet 1. Two independent raters (CK and MP) conducted a preliminary survey of the literature search results, evaluating both title and abstract for initial relevance. The two raters then reviewed the manuscripts of the screened publications for rigor prior to inclusion in the analysis. Inclusion criterion for manuscripts were: (1) an original research article, case report, case series, or trial of DBS co-occurring with pregnancy, (2) reporting on a DBS case in pregnancy that was carried out to completion (i.e., the subject had given birth). Exclusion criteria were: (1) duplicate publications, or conference proceeding of an eventual manuscript, (2) review articles or non-research articles or (3) did not include pregnancy information within the article. Published abstracts, letters to the editor or publications from conference proceedings were included in accordance with the recommendations put forth by Scherer and Saldanha (2019) if they met relevant pre-determined criteria. The inclusion and exclusion criteria used for manuscripts was applied to abstracts and letters to the editor. The following data were extracted from each article: the condition to treat, DBS target structure, maternal complications, route of delivery, birth complications, neonatal complications, neurodevelopmental follow-up, and other pertinent information offered. This review was made in accordance with PRISMA guidelines as presented in Figure 1 (Page et al., 2021).

A search of ClinicalTrials.gov for DBS studies involving dystonia, pain, epilepsy, MDD, obsessive-compulsive disorder (OCD), and Tourette syndrome (TS) was also conducted. The same two independent raters reviewed the results and included trials meeting the following criteria:(1) involved DBS of the relevant clinical condition and (2) offered DBS-based intervention. A trial was excluded if it (1) was a duplicate of another trial; (2) focused on another condition outside of the search; (3) did not involve DBS; or (4) was observational and not pertaining to a recently offered DBS based intervention. For all included trials, each rater reviewed the trial's inclusion and exclusion criteria. Three questions were addressed for each included trial. (1) Does the trial exclude subjects who are pregnant? (2) Does the trial exclude persons of childbearing age not on contraception deemed adequate by investigators and/or individuals intending to become pregnant? (3) Does the trial exclude breastfeeding individuals? Other data extracted from the clinical trial results included: enrollment status, regional location of trial, trial phase, date of trial start, and date of trial completion (if applicable).

# Results

## Deep brain stimulation

Results of the literature search are shown in **Table 1** and illustrated in **Figure 2**. Out of 681 publications, there were eight eligible publications reporting DBS in pregnancy that were included for analysis. All eight were full-length original research articles. These eight publications reported information on 27 patients with DBS who subsequently became pregnant and all received active neuromodulation treatment throughout their pregnancy (**Table 1**). Two individuals became pregnant twice with DBS for a total of 29 pregnancies. Another two persons conceived twin pregnancies, which brings the total to 31 infants. The most prevalent condition was dystonia (N = 17, 63%) followed by epilepsy (N = 4, 15%) and PD (N = 3, 11%). The most frequent target was the globus pallidus internus (N = 20, 74%) followed by the anterior nucleus of the thalamus (N = 4, 15%) and the subthalamic nucleus (N = 3, 11%).

Pregnancy complications for the 29 pregnancies were observed in three (12.9%) patients. Non-obstetric complications occurred in three individuals who had discomfort at the neurostimulator site (N = 2 subclavicular, N = 1 abdominal). One of these patients also had an obstetric complication

of a preterm delivery at 35 weeks gestational age. The discomfort experienced by three subjects was mechanically induced from physiologic abdominal and breast changes as pregnancy progressed. One additional patient was reported to have experienced the spontaneous abortion of one fetus in the early weeks of a twin pregnancy (Scelzo et al., 2015). The authors, however, determined that this adverse event was not related to DBS, thus it was not included in our complication data. As noted by Ziman et al. (2016) a subject who experienced abdominal discomfort from the neurostimulator site was the same individual who experienced preterm birth at 35 weeks. This patient had a battery readjustment procedure following her first pregnancy, relocating the battery from the abdomen to the subcutaneous tissue of the chest. With her subsequent second pregnancy, she experienced less discomfort and had a term delivery. Three publications did not include any followup data on neonatal outcomes. The remaining publications (N = 5) followed subjects from 6 to 108 months. No negative neurodevelopmental outcomes were reported.

No articles were found focusing specifically on DBS's effects on lactation or future fertility; however all subjects had a DBS stimulator prior to conception and reported no issues with becoming pregnant. Additionally, five out of eight publications followed DBS subjects from initiation of DBS treatment, into a pregnancy, and then into the postpartum period. These studies did not report any issues related to fertility. One subject reported inability to breastfeed due to discomfort. Otherwise, there were no reports in the literature of DBS directly affecting lactation.

Out of a potential of 277 clinical trials for the secondary objective, 135 trials on DBS met criteria (**Figure 1**). In clinical trials offering DBS lead implantation across all conditions, 68% (N = 92) excluded actively pregnant subjects, 22% (N = 29) excluded breastfeeding individuals, and 44% (N = 59) excluded persons of childbearing age intending to conceive within the coming years and/or individuals not on "adequate" contraception. Details on which contraceptive methods were deemed adequate were not specified in the exclusion criteria. Trials examining DBS for MDD were most likely to exclude the populations of interest (91%). The trial indication least likely to exclude this population was dystonia (41%) (**Table 2**).

## Discussion

Though our literature search for our primary objective to describe patients with DBS and pregnancy returned 681 potential articles, only eight met inclusion criteria for analysis. Data from these 8 reports of 29 pregnancies and 31 infants suggested reasonable safety profile for DBS in pregnancy. The 1 preterm birth (1 of 29 pregnancies, 4%) seems unlikely to be related to the DBS intervention since the expected rate of preterm birth in the general population would be approximately 10–12% (Walani, 2020). Discomfort at the battery site occurred



TABLE 1 Deep brain stimulation (DBS) literature review data.

Study (Author)	Subjects (N)	Infants (N)	Conditions* (N, %)	Target structure (N, %)	Pregnancy complications (N)	Follow-up (months)
Scelzo et al., 2015	11 <sup>a,b</sup>	13	PD (3), TS (2), OCD (1), dystonia (5)	Bilateral Gpi (8), STN (3)	Stimulator site discomfort (2)	24
Bóné et al., 2021	2	2	Epilepsy	ANT		40.5
Ziman et al., 2016	6 <sup>a</sup>	7	Dystonia	Bilateral Gpi	PTD at 35 weeks (1) + stimulator site discomfort (1)	50.8
Ozturk and Kadiroğulları, 2022	1	1	Dystonia	Bilateral Gpi		N/a
Paluzzi et al., 2006	3 <sup>b</sup>	4	Dystonia	Bilateral Gpi		N/a
Park et al., 2017	1	1	Dystonia	Bilateral Gpi		36
House et al., 2021	2	2	Epilepsy	ANT		15
Lefaucheur et al., 2015	1	1	Dystonia	Bilateral Gpi		N/a
Total	27	31 <sup>c</sup>	PD (3, 11%), TS (2, 7%), OCD (1, 4%), dystonia (17, 63%), epilepsy (4, 15%)	Bilateral GPi (20, 74%), STN (3, 11%), ANT (4, 15%)		Average (months) = 33.26

The table displays the results of the literature review for DBS in pregnancy, outlining the number of subjects/infants, conditions, target structures, and pregnancy complications for each of the included studies.

\*PD, Parkinson's disease; TS, Tourette's syndrome; OCD, obsessive-compulsive disorder; GPi, globus pallidus internus; STN, subthalamic nucleus; SAB, spontaneous abortion; ANT, anterior nucleus of thalamus; PTD, preterm delivery,  $a_n = 1$  subject became pregnant twice in this cohort,  $b_n = 1$  subject had a twin pregnancy in this cohort,  $c_n = 1/31$  fetus was spontaneously aborted in the first trimester determined to be unrelated to DBS.

in three pregnancies (10%) and may be a risk given the physiologic body changes related to pregnancy. Aside from the abdominal site, the nature of discomfort experienced by the

patients with pulse generators in the subclavicular region is curious. This may suggest migration of the DBS pulse generator in response to pregnancy related body habitus changes. BMI



### TABLE 2 Clinical trials review data.

	Condition	Included trials	Excludes pregnancy	Excludes POCBA*	Excludes breastfeeding
DBS					
	Pain	10	8 (80%)	6 (60%)	4 (40%)
	Dystonia	34	14 (41%)	4 (21%)	7 (21%)
	Tourette's	11	7 (64%)	3 (27%)	4 (36%)
	Epilepsy	15	11 (73%)	10 (67%)	2 (20%)
	Depression	35	32 (91%)	25 (71%)	6 (17%)
	Total (DBS)	135	92 (68%)	59 (44%)	29 (22%)

The table details the exclusion of various populations from clinical trials by modality of neuromodulation and the condition of interest for each trial.

\*Persons of childbearing age (POCBA) planning to conceive or not on "adequate" contraception.

was not discussed in the studies reporting neurostimulator site discomfort. All 27 patients had DBS leads implanted prior to pregnancy with no reported fertility issues across a variety of subcortical targets.

The explicit exclusion of pregnant individuals from 68% of interventional trials involving DBS is somewhat problematic. It is a legitimate concern to proceed with caution when there are questions whether there is adequate equipoise for including this vulnerable population given the paucity of data. Based on our review, there were no trials, including observational, specifically designed to study the safety or efficacy of DBS in the pregnant population. This impairs the ability to determine whether DBS could be a reasonable and effective intervention in pregnancy. Nevertheless, our data are reassuring that there are no clear significant safety signals as of yet. Importantly, there are data from other treatments demonstrating relative safety and effectiveness of interventions that use electrical stimulation, such as



cardioversion for arrhythmias and electroconvulsive therapy (ECT) in pregnancy (Yonkers et al., 2009; Enriquez et al., 2014). Interestingly, when reviewing the clinical trials for DBS in MDD, very specific inclusion criteria were utilized such that the pregnant and post-partum populations, of which there is a significant incidence and prevalence, were excluded One possible explanation for this could be the attempt to limit confounding variables given the historically variable response after DBS in the already very specific treatment resistant depression population.

It is unlikely that experience with DBS in only 29 pregnancies is sufficient to offer guidance on DBS and pregnancy. For pharmacotherapeutics, the Food and Drug Administration (FDA) typically requires data from clinical trials adequately powered before offering clinical guidance. Because DBS trials are designed to treat a specific indication, the only way to learn whether DBS should be a treatment option in pregnancy is to include (not exclude) pregnancy in treatment trials for those indications. This is consistent with the FDA's stance on research and pregnancy. In 2018 the FDA issued a draft of a guidance communicating there

were high exclusion rates of pregnant persons in clinical trials (Center for Drug Evaluation and Research, 2020). Guidance documents can be used to communicate regulatory hurdles and suggestions for industry to navigate the hurdles. This draft was revised in 2019, and has since seen no activity in revisions or suggestions. Specific medical devices were not mentioned in this document, which focused on pharmacologic agents. In the guidance, a framework was proposed by the FDA for inclusion of pregnant individuals in trials studying pharmacologic agents, which perhaps could be modeled to guide medical device trials in pregnancy. The FDA suggested the inclusion of pregnant individuals in clinical trials is ethically justifiable under specific criteria. These criteria, which differ slightly if the clinical trial is classified as pre- or post-market, are outlined in Figure 3.

There are several important issues to consider regarding potential for adverse effects of DBS in pregnancy. There are no animal models studying the safety of DBS in pregnancy and lactation or the effects of DBS on fertility. The first criterion for both pre-market and post-market studies from the FDA's draft for the inclusion of pregnant individuals in clinical trials thus has not been met. Furthermore, the mechanism of action of DBS is not fully delineated, making it difficult to anticipate potential downstream effects on the reproductive cycle (Herrington et al., 2016). There is a paucity of evidence that DBS may affect second messengers involved with reproductive physiology. For example, DBS of the nucleus accumbens (not a target structure in any study in our review) has been shown to result in shifts in prolactin and cortisol levels (de Koning et al., 2013, 2016). Another DBS target with potential reproductive consequences is the hypothalamus. DBS targeting the lateral hypothalamic nuclei has been used to treat obesity, cluster headache, generalized anxiety disorder, and post-traumatic stress-disorder (May, 2008; Franco et al., 2018; Li et al., 2022). There is potential for off-target stimulation of other hypothalamic nuclei. These off target effects could theoretically impact fertility, pregnancy, and lactation. Finally, pregnancy induces the production of a large amount of allopregnanolone, a steroid hormone with gamma aminobutyric acid (GABA)-ergic effects on the central nervous system. Given higher levels of GABA-like activity during pregnancy, pregnant individuals may theoretically require pregnancyspecific stimulation settings to successfully achieve a therapeutic effect (Kim et al., 2011).

When considering these issues, it may be helpful to examine the status of ECT in pregnancy as it is relevant to the discussion about DBS. ECT has been approved for use in pregnancy and there is little ongoing debate regarding its safety in this population. No animal model studies were provided to support the use of ECT in pregnancy. This safety determination was made jointly by amassing case reports in 1994, and the American College of Obstetricians and Gynecologists (ACOG) and the American Psychiatric Association confirmed the evidence in 2009 (Miller, 1994; Yonkers et al., 2009). A review by Rose et al found four meta-analyses studying ECT, three of which found similar numbers of cases in the literature of over 300 (Rose et al., 2020). The most recent (fourth) meta-analysis in 2014 removed cases before 1975, which marked the transition to modern ECT anesthesia away from the insulin coma. This eliminated confounding adverse effects, and yielded only 76 cases (Pompili et al., 2014). This process could serve as a framework for determining the safety of neuromodulation in pregnant or lactating individuals. We encourage obstetricians, neurologists, psychiatrists, and neurosurgeons to continue publishing more cases of DBS in pregnancy to strengthen the body of evidence.

The results of this comprehensive review should be interpreted with caution. Included studies were retrospective case reports or case series. Heterogeneity of cases, sample size, limited follow up, outcome reporting bias, and publication bias could all have impacted the results. Additionally, the reported neonatal and infant outcome metrics were inconsistently reported. The data collected on child neurodevelopment was heterogenous, with some data collected up to 108 months

and other studies failing to report any follow-up. Followup data, when available, varied from gross assessments to more specific milestone assessments conducted specifically by pediatricians. Further studies would be strengthened by uniform and detailed protocol assessments of neurodevelopmental progress. Nevertheless, the risk to the individual or fetus remains theoretical as the literature does not demonstrate evidence of harm beyond average obstetric risk. It is not theoretically feasible for electrical current from DBS stimulation to reach the uterus or developing fetus. There is little evidence to support the hypothetical risk of target stimulation leading to direct downstream effects on pregnancy, lactation, or fertility, similar to the use of ECT. The current clinical data for DBS in pregnancy are largely reassuring, demonstrate no clear adverse safety signals, and the sample size is large enough to justify inclusion of pregnant patients in well-designed clinical trials. Performing such trials is critically important to facilitate understanding of whether DBS has a role as an intervention in pregnancy and in reproductive age-patients. Without such studies, pregnant patients could be denied potentially effective treatment for serious conditions, which themselves could adversely impact obstetric and pediatric outcomes. Taking the complication rate of 12.9% in our cohort treated with DBS becomes particularly poignant when compared to, for example, a cohort of pregnant patients with dystonia without DBS experienced complications during pregnancy or delivery at a 45.26% rate (San Luciano et al., 2019). It is thus an imperative to find an inclusive set of criterion that allow for the access of a the latest treatment modalities to those most vulnerable.

# Conclusion

The data from 29 pregnancies in 27 subjects suggest that DBS during pregnancy does not have a high perinatal complication profile. The most common reported concern was device discomfort, which should be considered when planning device placement in individuals considering pregnancy. Many but not all clinical trials exclude pregnant individuals and the documentation of safety in larger numbers of subjects may make more clinical trials available for pregnant individuals in the future. Increasing the number of pregnancy-related publications will clarify the safety profile for individuals with DBS interested in becoming pregnant and those who may find out they are pregnant following DBS implantation. Though the safety profile is emerging, the still small number of cases has hampered regulatory agencies from offering clear guidance on safety and on inclusion of this population in clinical trials. It will be interesting to observe whether the guidance will be similar to that for ECT. A roadmap guiding investigators toward safe neuromodulation in pregnant and lactating individuals will be of utmost importance, as DBS

10.3389/fnhum.2022.997552

continues to expand indications, and other neuromodulation techniques gain popularity (e.g., transcranial magnetic stimulation and transcranial direct current stimulation).

# Data availability statement

The raw data supporting the conclusion of this article will be made available by the authors, without undue reservation.

# Author contributions

CK and TP contributed to drafting the manuscript and data extraction. CK, TP, KR-R, AR-Z, JS, MO, and JW contributed to analysis and interpretation of the data in their respective expertise. AR-Z, KR-R, JS, MO, and JW critically revised the manuscript. All authors approved the manuscript as submitted.

# Funding

JW's was supported by NIH R25NS108939. MO's was supported by: NIH R01 NR014852, R01NS096008, UH3NS119844, and U01NS119562 and also received the PI of the NIH R25NS108939 training grant.

# Conflict of interest

MO serves as Medical Advisor the Parkinson's Foundation, and had received research grants from NIH, Parkinson's Foundation, the Michael J. Fox Foundation, the Parkinson Alliance, Smallwood Foundation, the Bachmann-Strauss Foundation, the Tourette Syndrome Association, and the UF Foundation. MO's research was supported by: NIH R01 NR014852, R01NS096008, UH3NS119844, and U01NS119562.

References

Artama, M., Auvinen, A., Raudaskoski, T., Isojärvi, I., and Isojärvi, J. (2005). Antiepileptic drug use of women with epilepsy and congenital malformations in offspring. *Neurology* 64, 1874–1878. doi: 10.1212/01.WNL.0000163771.96 962.1F

Battino, D., and Tomson, T. (2007). Management of epilepsy during pregnancy. *Drugs* 67, 2727–2746. doi: 10.2165/00003495-200767180-00007

Bóné, B., Kovács, N., Balás, I., Horváth, R. A., Dóczi, T., and Janszky, J. (2021). Pregnancy and deep brain stimulation therapy for epilepsy. *Epilept. Disord.* 23, 633–638. doi: 10.1684/epd.2021.1304

CDC (2022). NVSS - Birth Data. Available online at: https://www.cdc.gov/nchs/ nvss/births.htm (accessed June 13, 2022).

Center for Drug Evaluation and Research (2020). Pregnant women: Scientific and ethical considerations for inclusion in clinical trials, U.S. Food and Drug Administration. FDA. Available online at: https://www.fda.gov/

MO had received royalties for publications with Demos, Manson, Amazon, Smashwords, Books4Patients, Perseus, Robert Rose, Oxford and Cambridge (movement disorders books). MO was an associate editor for New England Journal of Medicine Journal Watch Neurology and JAMA Neurology, and also participated in CME and educational activities (past 12-24 months) on movement disorders sponsored by WebMD/Medscape, RMEI Medical Education, American Academy of Neurology, Movement Disorders Society, and by Vanderbilt University. The institution and not MO receives grants from Medtronic, Abbvie, Boston Scientific, Abbott and Allergan, and the PI has no financial interest in these grants. MO had participated as a site PI and/or co-I for several NIH, foundation, and industry sponsored trials over the years but has not received honoraria. Research projects at the University of Florida receive device and drug donations.

The remaining authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

# Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

# Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fnhum. 2022.997552/full#supplementary-material

regulatory-information/search-fda-guidance-documents/pregnant-womenscientific-and-ethical-considerations-inclusion-clinical-trials (accessed June 30, 2022).

Chan, J., Natekar, A., Einarson, A., and Koren, G. (2014). Risks of untreated depression in pregnancy. *Can. Fam. Physician* 60, 242–243.

de Koning, P. P., Figee, M., Endert, E., Storosum, J. G., Fliers, E., and Denys, D. (2013). Deep brain stimulation for obsessive-compulsive disorder is associated with cortisol changes. *Psychoneuroendocrinology* 38, 1455–1459. doi: 10.1016/j. psyneuen.2012.12.006

de Koning, P. P., Figee, M., Endert, E., van den Munckhof, P., Schuurman, P. R., and Storosum, J. G. (2016). Rapid effects of deep brain stimulation reactivation on symptoms and neuroendocrine parameters in obsessive-compulsive disorder. *Transl. Psychiatry* 6:e722. doi: 10.1038/tp.2015.222 Denison, T., and Morrell, M. J. (2022). Neuromodulation in 2035: The neurology future forecasting series. *Neurology* 98, 65–72.

Enriquez, A. D., Economy, K. E., and Tedrow, U. B. (2014). Contemporary management of arrhythmias during pregnancy. *Circulation* 7, 961–967. doi: 10. 1161/CIRCEP.114.001517

Franco, R. R., Fonoff, E. T., Alvarenga, P. G., Alho, E. J. L., Lopes, A. C., Hoexter, M. Q., et al. (2018). Assessment of safety and outcome of lateral hypothalamic deep brain stimulation for obesity in a small series of patients with prader-willi syndrome. *JAMA Netw. Open* 1:e185275. doi: 10.1001/jamanetworkopen.2018. 5275

Harmsen, I. E., Elias, G. J. B., Beyn, M. E., Boutet, A., Pancholi, A., Germann, J., et al. (2020). Clinical trials for deep brain stimulation: Current state of affairs. *Brain Stimul.* 13, 378–385. doi: 10.1016/j.brs.2019.11.008

Herrington, T. M., Cheng, J. J., and Eskandar, E. N. (2016). Mechanisms of deep brain stimulation. J. Neurophysiol. 115, 19–38. doi: 10.1152/jn.00281.2015

House, P. M., Herzer, A., Lorenzi, I., Niedernhöfer, P., Voges, B., Stodieck, S., et al. (2021). Deep brain stimulation (DBS) of anterior nucleus thalami (ANT) in female epilepsy patients during pregnancy and delivery: Experience from two cases. *Epilept. Disord. Int. Epilepsy J. Videotape* 23, 933–936. doi: 10.1684/epd.2021. 1330

Kim, D. R., Epperson, N., Paré, E., Gonzalez, J. M., Parry, S., Thase, M. E., et al. (2011). An open label pilot study of transcranial magnetic stimulation for pregnant women with major depressive disorder. *J. Womens Health (2002)* 20, 255–261. doi: 10.1089/jwh.2010.2353

Kobau, R., Zahran, H., Thurman, D. J., Zack, M. M., Henry, T. R., Schachter, S. C., et al. (2008). Epilepsy surveillance among adults–19 States, Behavioral Risk Factor Surveillance System, 2005. *MMWR Surveill. Summ.* 57, 1–20.

Lefaucheur, R., Derrey, S., Borden, A., Verspyck, E., Tourrel, F., and Maltête, D. (2015). Patient with perinatal brain injury dystonia treated by deep brain stimulation: Management during pregnancy. *Rev. Neurol.* 171, 90–91. doi: 10.1016/j.neurol.2014.08.005

Li, H.-T., Donegan, D. C., Peleg-Raibstein, D., and Burdakov, D. (2022). Hypothalamic deep brain stimulation as a strategy to manage anxiety disorders. *Proc. Natl. Acad. Sci. U.S.A.* 119:e2113518119. doi: 10.1073/pnas.2113 518119

Louis, E. D., and Dogu, O. (2008). Does age of onset in essential tremor have a bimodal distribution? Data from a tertiary referral setting and a population-based study. *Neuroepidemiology* 29, 208–212. doi: 10.1159/000111584

May, A. (2008). Hypothalamic deep-brain stimulation: Target and potential mechanism for the treatment of cluster headache. *Cephalalgia* 28, 799–803. doi: 10.1111/j.1468-2982.2008.01629.x

Miller, L. J. (1994). Use of electroconvulsive therapy during pregnancy. *Hosp. Commun. Psychiatry* 45, 444–450. doi: 10.1176/ps.45.5.444

Ozturk, G., and Kadiroğulları, P. (2022). Management of pregnancy and childbirth in a cervical dystonia patient with an implanted deep brain stimulation system: A case report. *Ann. Indian Acad. Neurol.* 25, 121–123. doi: 10.4103/aian. AIAN\_151\_21

Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D., et al. (2021). The PRISMA 2020 statement: An updated guideline for reporting systematic reviews. *BMJ* 372:n71.

Paluzzi, A., Bain, P. G., Liu, X., Yianni, J., Kumarendran, K., and Aziz, T. Z. (2006). Pregnancy in dystonic women with in situ deep brain stimulators. *Mov. Disord.* 21, 695–698. doi: 10.1002/mds.20777

Park, H. R., Lee, J. M., Park, H., Shin, C. W., Kim, H. J., Park, H. P., et al. (2017). Pregnancy and delivery in a generalized dystonia patient treated with internal globus pallidal deep brain stimulation: A case report. *J. Korean Med. Sci.* 32, 155–159. doi: 10.3346/jkms.2017.32.1.155

Pompili, M., Dominici, G., Giordano, G., Longo, L., Serafini, G., Lester, D., et al. (2014). Electroconvulsive treatment during pregnancy: A systematic review. *Expert Rev. Neurother.* 14, 1377–1390. doi: 10.1586/14737175.2014.972373

Rose, S., Dotters-Katz, S. K., and Kuller, J. A. (2020). Electroconvulsive therapy in pregnancy: Safety, best practices, and barriers to care. *Obstet. Gynecol. Surv.* 75, 199–203. doi: 10.1097/OGX.00000000000763

San Luciano, M., Shanker, V., Bressman, S., Raymond, D., and Saunders-Pullman, R. (2019). Pregnancy and delivery complications in women with inherited isolated dystonia [abstract]. *Mov. Disord.* 34(Suppl. 2).

Sazgar, M. (2019). Treatment of women with epilepsy. Continuum (Minneapolis, Minn.) 25, 408–430. doi: 10.1212/CON.00000000000713

Scelzo, E., Mehrkens, J. H., Bötzel, K., Krack, P., Mendes, A., Chabardès, S., et al. (2015). Deep brain stimulation during pregnancy and delivery: Experience from a series of "DBS Babies". *Front. Neurol.* 6:191. doi: 10.3389/fneur.2015.00191

Scherer, R. W., and Saldanha, I. J. (2019). How should systematic reviewers handle conference abstracts? A view from the trenches. *Syst. Rev.* 8:264. doi: 10.1186/s13643-019-1188-0

Schrag, A., and Schott, J. M. (2006). Epidemiological, clinical, and genetic characteristics of early-onset parkinsonism. *Lancet Neurol.* 5, 355–363. doi: 10. 1016/S1474-4422(06)70411-2

Van Niel, M. S., and Payne, J. L. (2020). Perinatal depression: A review. *Cleve. Clin. J. Med.* 87, 273–277. doi: 10.3949/ccjm.87a.19054

Vitturi, B. K., Cabral, F. B., and Cukiert, C. M. (2019). Outcomes of pregnant women with refractory epilepsy. *Seizure Eur. J. Epilepsy* 69, 251–257. doi: 10.1016/j.seizure.2019.05.009

Walani, S. R. (2020). Global burden of preterm birth. Int. J. Gynaecol. Obstet. 150, 31–33. doi: 10.1002/ijgo.13195

Yonkers, K. A., Wisner, K. L., Stewart, D. E., Oberlander, T. F., Dell, D. L., Stotland, N., et al. (2009). The management of depression during pregnancy: A report from the American Psychiatric Association and the American College of Obstetricians and Gynecologists. *Obstet. Gynecol.* 114, 703–713. doi: 10.1097/ AOG.0b013e3181ba0632

Ziman, N., Coleman, R. R., Starr, P. A., Volz, M., Marks, W. J. Jr., Walker, H. C., et al. (2016). Pregnancy in a series of dystonia patients treated with deep brain stimulation: Outcomes and management recommendations. *Stereotact. Funct. Neurosurg.* 94, 60–65. doi: 10.1159/000444266